Author's response to reviews

Title: Conducting Online Expert Panels: A Feasibility and Experimental Replicability Study

Authors:

Dmitry Khodyakov (dkhodyak@rand.org)
Susanne Hempel (Susanne_Hempel@rand.org)
Lisa Rubenstein (Lisa.Rubenstein@va.gov)
Paul Shekelle (Shekelle@rand.org)
Robbie Foy (R.Foy@leeds.ac.uk)
Susanne Salem-Schatz (SSS@hcqi.com)
Sean O'Neill (Sean.M.Oneill@gmail.com)
Margie Danz (MJSDanz@gmail.com)
Siddhartha Dalal (SDalal@rand.org)

Version: 5 Date: 20 December 2011

Author's response to reviews:

Hello,

Please see the attached. If you think any other changes are required, I would appreciate if you could take of them yourselves.

Thanks,

Dmitry Khodyakov